

Dear Reviewer,

Thank you very much for your thoughtful and detailed review of my manuscript. I sincerely appreciate your constructive feedback, which will help me further improve the quality of the paper. I will address each of your comments carefully to ensure the manuscript meets the highest standards.

Reviewer 1,

**Comment 1:**

Keywords: The reviewer suggests reducing the number of keywords to five.

**Response 1:**

I will reduce the number of keywords to five as per your suggestion. This will help focus on the most essential terms that capture the core of the study, ensuring better clarity and precision. Keywords: Non-suicidal self-injury, adolescent, network analysis, latent profile analysis

**Comment 2:**

In introduction, the content is obtrusive: Over 80% of individuals..... It is suggested that this paragraph be integrated with the previous paragraph. " seems obtrusive and should be integrated with the previous paragraph for better flow.

**Response 2:**

Introduction: Thank you for pointing this out. I have already revised the introduction and integrated the "Over 80% of individuals..." paragraph with the preceding section to ensure better flow and coherence in the narrative. If needed, I am happy to upload the revised manuscript for your review.

**Comment 3:**

References: Updating the references with the latest research is recommended.

**Response 3:**

Thank you for the suggestion. We have thoroughly reviewed and updated the references in accordance with your recommendation. Several new references, reflecting the latest research in the field, have been added, particularly in the Discussion section to support and expand upon the revised content.

**Comment 4:**

Network Analysis Code: The reviewer requests the complete code for verification purposes.

**Response 4:**

Thank you for your request. We are happy to provide the complete R code used in our network analysis to ensure transparency and facilitate verification. Below, we outline the analysis steps and include the corresponding code.

Overview of the Analysis:

We employed network analysis to explore the interactions between risk and resilience factors across the identified NSSI subgroups. The analysis involved the following steps:

**Network Estimation**

We estimated Gaussian Graphical Models (GGMs) for each subgroup using the qgraph package (Epskamp et al., 2012). GGMs capture conditional dependencies between variables by estimating partial correlations. We applied the graphical Least Absolute Shrinkage and Selection Operator (LASSO) regularization (Friedman et al., 2008) to prevent overfitting and produce sparse networks.

**Network Inference (Centrality Measures)**

Centrality measures were calculated to assess the importance of each node (variable) within the network:

Node Strength: Indicates how strongly a node is directly connected to others.

Closeness Centrality: Reflects how close a node is to all other nodes based on the shortest paths.

**Betweenness Centrality:** Measures the extent to which a node lies on the shortest paths between other nodes.

**Network Stability and Accuracy**

We assessed the stability and accuracy of the network estimates using the bootnet package (Epskamp et al., 2018). Bootstrapped confidence intervals were generated for edge weights and centrality measures through non-parametric bootstrapping (2,000 bootstraps). We calculated the correlation stability coefficient (CS-coefficient) for centrality measures, with values above 0.5 indicating high stability.

## **Network Comparison**

The Network Comparison Test (NCT) from the NetworkComparisonTest package (van Borkulo et al., 2017) was used to statistically compare the network structures between the NSSI subgroups. We performed permutation tests to assess differences in global strength and specific edge weights, applying the Holm-Bonferroni correction for multiple comparisons.

## **Network Density**

We calculated the density of each network, defined as the average strength of all edges, to compare overall connectivity between subgroups (Burger et al., 2023).

## **Exploratory and Confirmatory Factor Analysis**

Exploratory Factor Analysis (EFA) and Confirmatory Factor Analysis (CFA) were conducted to further understand the underlying structure of the data.

## **R Code for the Analysis:**

```
# 1. Load Required Packages
```

```
library(psych)
```

```
library(GPArotation)
```

```
library(lavaan)
```

```
library(dplyr)
```

```
library(qgraph)
```

```
library(bootnet)
```

```
library(NetworkComparisonTest)
```

```
# 2. Data Loading and Preparation
```

```
# Load your dataset (replace 'your_data_file.csv' with your actual data file)
```

```
# data <- read.csv("your_data_file.csv")
```

```
# Prepare the data frame with relevant variables
```

```
dataYN <- data.frame(
```

```
  yts_1 = data$yts_1,
```

```
  yts_2 = data$yts_2,
```

```
  yts_3 = data$yts_3,
```

```
  yts_4 = data$yts_4,
```

```

yts_5 = data$yts_5,
yts_6 = data$yts_6,
yts_7 = data$yts_7,
yts_8 = data$yts_8,
yts_9 = data$yts_9,
yts_10 = data$yts_10,
yts_11 = data$yts_11,
yts_12 = data$yts_12,
yts_13 = data$yts_13
)
# 3. Network Estimation
# Estimate the network using EBICglasso
network_estimation <- estimateNetwork(dataYN, default = "EBICglasso")
# Plot the estimated network
qgraph(network_estimation$graph, layout = "spring", title = "Estimated Network")
# 4. Network Inference (Centrality Measures)
# Calculate centrality measures
centrality_measures <- centrality(network_estimation)
# Plot centrality measures
centralityPlot(network_estimation)
# 5. Network Stability and Accuracy (Bootstrapping)
# Assess the stability of the network
boot_results <- bootnet(network_estimation, nBoots = 2000, statistics = c("edge",
"strength"))
# Plot bootstrapped edge weights
plot(boot_results, "edge")
# Plot bootstrapped strength centrality
plot(boot_results, "strength")
# Calculate the correlation stability coefficient
cs_coeff <- corStability(boot_results)

```

```

print(cs_coeff)

# 6. Network Comparison Test

# Assuming 'data_group1' and 'data_group2' are data frames for different NSSI
subgroups

# Replace with your actual group data
# data_group1 <- subset(dataYN, group == 1)
# data_group2 <- subset(dataYN, group == 2)

# Perform the Network Comparison Test
# nct_results <- NCT(data_group1, data_group2, it = 1000)

# Summarize and plot NCT results
# summary(nct_results)
# plot(nct_results)

# 7. Network Density

# Calculate network density for each subgroup
#
#           density_group1
#           <-
# mean(abs(network_group1$graph[upper.tri(network_group1$graph)]))
#
#           density_group2
#           <-
# mean(abs(network_group2$graph[upper.tri(network_group2$graph)]))

# 8. Exploratory Factor Analysis (EFA)

# Compute the polychoric correlation matrix
R1 <- polychoric(dataYN)$rho

# Determine the number of factors using parallel analysis
fa.parallel(R1, n.obs = nrow(dataYN), fa = "fa", main = "Parallel Analysis Scree
Plots")

# Perform factor analysis
fa_results <- fa(R1, nfactors = 1, fm = "wls", rotate = "varimax")
print(fa_results, digits = 2)

# 9. Confirmatory Factor Analysis (CFA)

# Prepare data for CFA

```

```

dataCFA1 <- na.omit(cbind(data$Id, dataYN))
colnames(dataCFA1)[1] <- "id"
# Specify the CFA model
model1 <- '
  EA =~ yts_1 + yts_2 + yts_3 + yts_4 + yts_5 + yts_6 + yts_7 + yts_8 + yts_9 +
yts_10 + yts_11 + yts_12 + yts_13
# Fit the CFA model
fit1 <- cfa(
  model1,
  data = dataCFA1,
  meanstructure = TRUE,
  std.lv = TRUE,
  estimator = "WLSMV",
  ordered = paste0("yts_", 1:13)
)
# Summarize CFA results
summary(fit1, fit.measures = TRUE, standardized = TRUE)
# Optional: Get factor scores
dataCFA1$factor1 <- lavPredict(fit1)
# Merge factor scores back into the main dataset
data <- left_join(data, dataCFA1[, c("id", "factor1")], by = "id")

```

**Comment 5:**

Formatting: There are some inconsistencies, such as line spacing, that should be addressed.

**Response 5:**

I appreciate your attention to detail. I will carefully review the manuscript to correct any formatting inconsistencies, such as line spacing, to ensure it adheres to the publication standards.

**Comment 6:**

Discussion Depth: The discussion section should be deepened to explore the results

more thoroughly.

**Response 6:**

Thank you for your feedback. I have already revised the manuscript, including a comprehensive update to the discussion section, to provide a more in-depth exploration of the results. I believe the current version addresses your concerns, but I am happy to make further revisions if necessary.

**Comment 7:**

There is a question that the author needs to explain to me. In the last paragraph of 4.5 in the discussion, does the revised discussion mean that there has been a round of revisions? If so, since I have not seen the initial version, could you briefly explain to me the difference between the initial version and the revised version.

**Response 7:**

Thank you for your insightful comments and for pointing out the issue in the last paragraph of section 4.5 in the discussion. We apologize for any confusion caused.

The mention of "our revised discussion reflects a deeper engagement with the reviewer's insights" was inadvertently included in the manuscript. This statement was intended for our internal revision notes and should have been omitted from the submitted version. There has not been a previous round of revisions shared with reviewers.

To clarify, we have now removed this paragraph from the manuscript to prevent any misunderstanding. The discussion section has been thoroughly revised to deepen the analysis of our study's findings, as per your recommendation. We have expanded on the key findings, explored theoretical and clinical implications, considered cultural contexts, acknowledged limitations, and suggested future research directions. These revisions aim to provide a more comprehensive and in-depth discussion of the results.

We appreciate your careful review and the opportunity to improve our manuscript.

Thank you again for your contribution to this study and hope to reply as soon as possible. To sum up, I think the author needs to respond to the above requirements and make minor modifications. Best wishes